Author’s response to reviews

Title: Can Body Mass Index influence the fracture zone in the fifth metatarsal base? A retrospective review

Authors:

Mattia Pugliese (mattiapugliese@outlook.com)
Daniele De Meo (dannydemeo@hotmail.com)
Ennio Sinno (enniosinno@gmail.com)
Virginia Pambianco (v.pambianco@hotmail.com)
Armando Cavallo (armandocavallo90@gmail.com)
Pietro Persiani (pietro.persiani@uniroma1.it)
Ciro Villani (ciro.villani@uniroma1.it)

Version: 4 Date: 16 Jan 2020

Author’s response to reviews:

You were previously asked if your statistical analysis allowed you to determine if a relationship exists, or not. This does not appear to have been corrected (see abstract and lines 41-42), despite you reporting that it had. Please undertake a careful review of all comments previously raised by the reviewers and confirm that appropriate adjustment has been made.

Manuscript and abstract were adjusted as per Editor’s suggestion

Lines 100 - 103 (discussion) should this sentence be incorporated later in the discussion (e.g. from line 118).

This portion of the manuscript was modified as per Editor’s suggestion